Author's response to reviews

Title: Preventing childhood obesity during infancy in UK primary care: a mixed-methods study of HCPs' knowledge, beliefs and practice

Authors:

Sarah A Redsell (Sarah.Redsell@nottingham.ac.uk)
Philippa J Atkinson (p.j.atkinson@bopenworld.com)
Dilip Nathan (dilip.nathan@nhs.net)
A Niroshan Siriwardena (nsiriwardena@lincoln.ac.uk)
Judy A Swift (judy.swift@nottingham.ac.uk)
Cristine Glazebrook (cris.glazebrook@nottingham.ac.uk)

Version: 4 Date: 16 June 2011

Author's response to reviews:

Hi there

Thanks for sending this paper back to me for amendments. I think I have done all you asked. My only query is that you asked for each affiliation to be written in full. My understanding is that they were already. Two of the authors are clinical staff and do not have standard academic addresses (Atkinson and Nathan).

Let me know if there is anything more you want me to do.

Best wishes

Sarah